---
title: "Integrations: Introduction"
sidebarTitle: "Introduction"
description: "Integrations make it easy to authenticate and use APIs."
---

<CardGroup>
  <Card
    title="What are Integrations?"
    icon="circle-question"
    href="/documentation/concepts/integrations"
  >
    An explanation of what they are and how they work
  </Card>
  <Card
    title="How to use Integrations"
    icon="gears"
    href="/documentation/guides/using-integrations"
  >
    How to connect Integrations to your account and use them in your code
  </Card>
  <Card title="How to create your own Integrations" icon="paintbrush" href="/integrations/create">
    You can create your own Integrations (and optionally share them with others)
  </Card>
  <Card title="Example repo & Jobs" icon="code" href="/examples/introduction">
    Check out example Jobs using our integrations.{" "}
  </Card>
</CardGroup>

## The Integrations Catalog

Navigate the menu or select Integrations from the table below.

| API                                       | Description                                                      | Webhooks | Tasks |
| ----------------------------------------- | ---------------------------------------------------------------- | -------- | ----- |
| [Airtable](/integrations/apis/airtable)   | Interact with the Airtable API                                   | 🕘       | ✅    |
| [GitHub](/integrations/apis/github)       | Subscribe to webhooks and perform actions                        | ✅       | ✅    |
| [Linear](/integrations/apis/linear)       | Streamline project and issue tracking                            | ✅       | ✅    |
| [OpenAI](/integrations/apis/openai)       | Generate text and images. Including longer than 30s prompts      | N/A      | ✅    |
| [Plain](/integrations/apis/plain)         | Perform customer support using Plain                             | 🕘       | ✅    |
| [Replicate](/integrations/apis/replicate) | Run machine learning tasks easily at scale                       | N/A      | ✅    |
| [Resend](/integrations/apis/resend)       | Send emails using Resend                                         | 🕘       | ✅    |
| [SendGrid](/integrations/apis/sendgrid)   | Send emails using SendGrid                                       | 🕘       | ✅    |
| [Slack](/integrations/apis/slack)         | Send Slack messages                                              | 🕘       | ✅    |
| [Stripe](/integrations/apis/stripe)       | Interact with the Stripe API                                     | ✅       | ✅    |
| [Supabase](/integrations/apis/supabase)   | Interact with your projects and databases                        | ✅       | ✅    |
| [Typeform](/integrations/apis/typeform)   | Interact with the Typeform API and get notified of new responses | ✅       | ✅    |
